Author's response to reviews

Title: Appropriate health-seeking behavior and associated factors among people who had cough for at least two weeks in northwest Ethiopia: A population-based cross-sectional study

Authors:

Meseret Senbeto (messi_97@yahoo.com)
Sebsibe Tadesse (sbsbtadesse90@gmail.com)
Takele Tadesse (takeletedesse1627@gmail.com)
Tesfahun Melesse (tesfahunmelesse@yahoo.com)

Version: 8 Date: 17 December 2013

Author's response to reviews: see over
To: Editorial Office of BMC Public Health

Ref: MS: 1749143240915571

Dear editor-in-chief,

Thank you for showing interest on our manuscript entitled “Appropriate health-seeking behavior and associated factors among people who had cough for at least two weeks in northwest Ethiopia: A population-based cross-sectional study”. We also appreciate the reviewers for their critical evaluation and comments. We have responded to those comments point-by-point (attached). We did also extensive revision and editing in order to fulfill the critical questions raised. Hope the response we have provided and the changes we have made would satisfy the reviewer and the Editor.

Thank you so much for your attention.

Kind regards:

Sebsibe Tadesse
The Corresponding Author
Editor’s Comments and Authors’ response

<table>
<thead>
<tr>
<th>Editor’s Comments</th>
<th>Authors’ response</th>
</tr>
</thead>
<tbody>
<tr>
<td>Thank you so much for showing interest on our manuscript!</td>
<td>It is excellent observation.</td>
</tr>
</tbody>
</table>
| "It appears that the reviewers are satisfied with the article and that:  
  • The one outstanding issue of other references of studies in Ethiopia are a moot point, except for the fact that there are two somewhat relevant recent references the authors may want to review and include.  
  • Also reference 23 needs to reflect authors’ last names - these show first names, John, Janet etc. |  
  • We have removed reference 7 from paragraph 1 of the discussion section.  
  • Sorry. We have modified reference 23. |
| Thank you so much! | }